Author's response to reviews

Title: Associations of Military Divorce with Mental, Behavioral, and Physical Health Outcomes

Authors:

Lawrence C Wang (lawrence.wang@med.navy.mil)
Amber Seelig (Amber.Seelig@va.gov)
Shelley MacDermid Wadsworth (Shelley@purdue.edu)
Hope McMaster (Hope_McMaster@abtassoc.com)
John E Alcaraz (jalcaraz@mail.sdsu.edu)
Nancy Crum-Cianflone (nancy32red@yahoo.com)

Version: 2
Date: 28 August 2014

Author's response to reviews:

I have revised the manuscript to comply with the following requests:

1. Format the abstract according to BMC guidelines.
2. Include a 'Competing Interests' declaration.
3. Add line numbers in the manuscript.